Author's response to reviews

Title: Implementation of a population-based epidemiological rare disease registry: Study protocol of the amyotrophic lateral sclerosis (ALS) - registry Swabia

Authors:

Gabriele Nagel (gabriele.nagel@uni-ulm.de)
Hatice Uenal (hatice.uenal@uni-ulm.de)
Angela Rosenbohm (la.rosenbohm@uni-ulm.de)
Albert C Ludolph (albert.ludolph@rku.de)
Dietrich Rothenbacher (dietrich.rothenbacher@uni-ulm.de)

Version: 3 Date: 14 December 2012

Author's response to reviews: see over
Dear Mr Silvestre,

Thank you for the opportunity to resubmit our manuscript “Implementation of a population-based epidemiological rare disease registry: Study protocol and first results of the amyotrophic lateral sclerosis (ALS) - registry Swabia” to BMC Neurology.

According to the editorial email
1) We have removed all results except the numbers and percentages of men and women in the study.
2) We have removed all baseline results from specifically the discussion and conclusion.
3) We now comment on the stage of our study in the material and method section.

All authors of this research paper have directly participated in the planning, execution, or analysis of the study. The authors have read and approved this manuscript. This manuscript is currently not submitted for press at any other journal.

We look forward to a favorable consideration of our manuscript and remain

Yours sincerely,

Gabriele Nagel